Author's response to reviews

Title: Effect of Payena dasyphylla on hyaluronidase enzyme activity and MMPs expression in IL-1beta stimulated human chondrocyte cell line (NHAC-kn)

Authors:

Kamini Citalingam (kaminicitalingam@gmail.com)
Seema Zareen (seema_zareen@yahoo.com)
Shamsul Khamis (shamsul@ibs.upm.edu.my)
Khozirah Shaari (khozirah@yahoo.com.my)
Syahida Ahmad (syahida@biotech.upm.edu.my)

Version: 2 Date: 18 July 2012

Author's response to reviews: see over
AUTHOR’S RESPONSE TO EDITORIAL’S COMMENTS ON RESEARCH ARTICLE (MS: 1228483942761486)

Dear Ms Carmela S. Quidoles,

Our research groups have checked the article [MS: 1228483942761486] that have been sent back to us and have made several corrections on the article.

Below are the corrections made on the manuscript based on the editorial requests:

- The author has included the competing interests section in the manuscript.
- The author has made changes on the abstract in accordance to the journal’s guidelines.
- The structure of the manuscript has been adjusted according to the journal’s requirement.
- The figures and tables section has been uploaded separately into the submission system.

Thank you for your time and consideration.

Yours sincerely,

Syahida Ahmad